Author's response to reviews

Title: P300 amplitude is insensitive to working memory load in schizophrenia

Authors:

   Pablo A Gaspar (pgaspar@neuro.med.uchile.cl)
   Sergio Ruiz (smruiz@med.puc.cl)
   Francisco Zamorano (fjzamora@puc.cl)
   Marcela Altayó (maltayo@puc.cl)
   Carolina Pérez (carolinaperezjara@gmail.com)
   Conrado A Bosman (c.bosman@donders.ru.nl)
   Francisco Aboitiz (faboitiz@puc.cl)

Version: 6 Date: 9 February 2011

Author's response to reviews: see over
Dear Sirs.
The BMC Series Editorial Production Team

We are glad to hear that the editorial board considers our article suitable for publication in the journal.

Herein we submit the revised version of our article, now entitled “P300 amplitude is insensitive to working memory load in schizophrenia” to BMC Psychiatry. We have addressed all concerns raised by Editorial Production Team.

Sincerely yours,

Pablo Gaspar
Q: Tables: you have uploaded the tables as figure files. Please remove them from the submission system and include the tables within the text file of the manuscript following the references. The tables should not be embedded images or drawings and should be formatted from cells, not tabbed text. Please also move the table title to above the table and the legend to below the table, within the text.

A: We have included the tables 1 and 2 within the manuscript as it was requested.

Q: Affiliations: Please remember to include the country in the affiliation details.

A: We have included the country in all affiliations.

Q: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

A: We have carefully revised our manuscript to avoid typographical errors.